Check for updates

#### **OPEN ACCESS**

EDITED BY Adriano Bisello, Eurac Research, Italy

REVIEWED BY Richard Kotter, Northumbria University, United Kingdom Jake Barnes, University of Oxford, United Kingdom

\*CORRESPONDENCE David G. Barns ⊠ d.g.barns@leeds.ac.uk

RECEIVED 16 November 2023 ACCEPTED 02 April 2024 PUBLISHED 16 May 2024

#### CITATION

Barns DG, Bale CSE, Taylor PG and Owen A (2024) Heat and the planning system: how can local authorities encourage deployment of low and zero-carbon heating? *Front. Sustain. Cities* 6:1339709. doi: 10.3389/frsc.2024.1339709

#### COPYRIGHT

© 2024 Barns, Bale, Taylor and Owen. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# Heat and the planning system: how can local authorities encourage deployment of low and zero-carbon heating?

David G. Barns<sup>1\*</sup>, Catherine S. E. Bale<sup>1,2</sup>, Peter G. Taylor<sup>1,2</sup> and Alice Owen<sup>1</sup>

<sup>1</sup>School of Earth & Environment, Faculty of Environment, University of Leeds, Leeds, United Kingdom, <sup>2</sup>School of Chemical & Process Engineering, Faculty of Engineering & Physical Sciences, University of Leeds, Leeds, United Kingdom

There is widespread recognition of the need for new homes to feature only low or zero-carbon (LZC) heating. However, residential developers continue to choose conventional high-carbon options such as natural gas boilers over net-zero compatible alternatives. This study explores how UK local authorities (LAs) within the English planning system can encourage residential developers to deploy LZC heating systems within their projects. We adopt an embedded case study design and analyse 30 residential project proposals within two LA areas. Our study examines local planning policies and interactions between developers and LA officers, along with the resultant outcomes, through documentary analysis and expert interviews with local stakeholders. We find that LAs can encourage developers to adopt LZC heating technologies above and beyond what is required nationally. The conditions for this to occur are (1) a planning policy which restricts allowable heating technology options, (2) empowering LA officers to enforce policies, (3) advice and support for developers to consider alternatives, and where necessary, (4) political backing to challenge unwilling developers. Study findings highlight the important role of LAs in creating the conditions for the transition to LZC heating, which show how they can use powers within the planning system to encourage developers to make low carbon choices without the need for direct investment.

#### KEYWORDS

heat decarbonisation, urban energy transition, real-estate developers, planning policy and practice, local authorities

# **1** Introduction

As one of 38 countries responsible for 25% of global emissions (Ritchie et al., 2023; ECIU, 2024), the UK has enshrined commitments in law to achieve net-zero carbon emissions by 2050 [The Climate Change Act, 2008 (2050 Target Amendment) Order 2019, 2019]. Globally, buildings are responsible for 37% of greenhouse gas (GHG) emissions (IEA, 2020), with 11.5% directly attributable to homes (Ge et al., 2022). Net-zero targets cannot be met without addressing how our homes and buildings are heated. The challenge of decarbonising the residential sector is illustrated in the UK, where 86% of the country's 29 million homes are heated by natural gas boilers and 1.7 million new boilers are installed every year (BEIS, 2018; MHCLG, 2021a,b). At the current rate of progress in the UK it would take many hundreds of years to replace fossil-fuel heating with low and zero-carbon (LZC) heating alternatives and achieve net-zero emissions from buildings (Rosenow and Thomas, 2020; Rosenow et al., 2020; Shabha et al., 2021). Unlike the electricity system, which can largely be transitioned from high- to low-carbon intensity remotely with little noticeable impact on the end user, heat is typically generated at or near the point of use and tends to

be subject to individual household decision-making (National Grid, 2018; Knobloch et al., 2019). Heating transitions in the residential sector are further complicated by the range of tenure types (owneroccupier, private rented, social rented, etc.) within the same area, such as a residential street or even a building (a tower block, for example). This causes challenges for those seeking to take an area or neighbourhood approach or encourage the deployment of shared heating technologies such as heat networks.

In this study, we focus on LZC heating deployment in new residential developments. Whilst the majority of emissions will need to be addressed in homes and buildings already constructed, the UK will likely see eight million new homes by 2050, representing a significant opportunity to implement LZC heating (CCC, 2020). The UK government and their scientific advisors recognise new homes as ideal settings to deploy less-established LZC heat technologies because, for example, they can be designed to accommodate different-sized heat emitters and space for thermal storage (CCC, 2019; BEIS, 2021). This approach will, in turn, help build a supply chain and develop skills across the workforce (Clarke et al., 2020; BEIS, 2021). New homes also offer a single point of decision-making during project development, which may well no longer exist following construction and sale.

Whilst the study applies to a range of technologies, we find particular emphasis within the results on shared ground loops (SGLs) as a suitable LZC heating option. SGLs remain a niche technology, and they have been shown to offer significant opportunities to help decarbonise the UK's housing stock, especially in homes that would struggle to accommodate their own air source heat pump (ASHP) or that are out of the reach of a heat network (Howard and Crook, 2021; Bale et al., 2022; Barns, 2022). SGLs are known by several names, but importantly combine aspects of heat pumps and heat networks and involve distributed heat pumps connected to a shared ground array through an ambient temperature heat network (GLA Etude, 2018; Boesten et al., 2019; Buffa et al., 2019; Lund et al., 2021; Ofgem, 2021a; BEIS Triple Point Heat Networks, 2022; Nesta, 2022; Element Energy, 2023).

The shift to LZC heating technologies in urban settings takes place within a broader socio-political regime (Webb et al., 2016, 2017; Tingey and Webb, 2020). This can impact decisions by those seeking to support or challenge the transition. The regime for residential heating in the UK features lock-in characteristics such as a well-established natural-gas grid serving most UK homes (Dodds and McDowall, 2013; Gross and Hanna, 2019); uncertainty around a shift to electrification vs. a repurposed gas grid to carry hydrogen for residential heating (Lowes et al., 2020b); a large gap between electricity and gas price that makes electrified options less favourable in operating costs (Barnes and Bhagavathy, 2020); a national building regulations policy which has not recognised the decarbonising of electricity grids in prescribed carbon factors (Rees, 2019); and a generally poor housing stock which aligns well with the instant heat and high temperatures of fossil heating, but is very inefficient (Chaudry et al., 2015; Guertler et al., 2015).

Given a challenging UK socio-political regime, the role of a few local authorities (LAs) in creating the conditions for low-carbon heating alternatives to flourish in cities has received significant attention (Bale et al., 2012; Schwanen, 2015; Bush et al., 2016; Fudge et al., 2016; Webb et al., 2017; Adam, 2018; Tingey and Webb, 2020). LAs can potentially take both active and passive policy measures. For example, if they have their own social housing stock, LAs can invest in retrofitting energy-efficient and low-carbon technologies. This has been underway, for instance, in Leeds, where Leeds City Council (LCC) has been connecting LA-owned homes to a city centre district heat network, installing SGL systems, and carrying out other energy efficiency improvements (LCC, 2021). Active investments in LZC measures of this type are, however, subject to the context of severe budget cuts since the early 2010s, which restricts the capacity of many UK local authorities to invest in similar active measures (Gray and Barford, 2018).

Most research about the role of LAs in supporting the energy transition has focused on exploring and evaluating active or strategic measures that typically involve the LA taking a lead role in developing and/or owning low-carbon infrastructure (Sullivan et al., 2013; Fudge et al., 2016; Bush et al., 2017; Webb et al., 2017; Roelich et al., 2018; Tingey and Webb, 2020). The strategic energy planning undertaken by empowered city authorities, especially in Denmark, has also received research attention (Sperling et al., 2011; Thellufsen and Lund, 2016; Krog, 2019). Yet, there has to date been less research focused on the ability of local authorities to encourage developers to choose LZC heating options through the planning system. Local authorities in the UK fulfil statutory responsibilities to develop and implement local spatial planning policies, thus providing both a "stick" (setting and enforcing construction standards) and a "carrot" (designating land for development) (MHCLG, 2019, 2021a; Ellis and Fieth, 2021; Ellis, 2022). This may mean LAs can deliver local decarbonisation objectives without the need for direct investment.

We explored the ability of LAs to shape developer technology decisions in new building settings within the UK's socio-political regime through the following research question:<sup>1</sup>

How can local authorities in England use the planning system to encourage residential developers to deploy low and zero-carbon heating systems in new homes?

Section 2 establishes the socio-technical transitions literature that underpins the analysis in this study. Section 3 details the methods used to address the research question through an embedded case study of two large UK cities across a sample of 30 residential developments. Section 4 presents the results of the case study, which finds that through implementing a set of local conditions, LAs can successfully shape the local domestic heat regime and encourage developers to choose LZC heating in their projects. Section 5 further discusses the implications of the findings on the potential for LAs to shape heat decarbonisation in their area. Section 6 concludes and outlines how the findings, whilst being UK-focused, could still be applied wherever local state actors seek to encourage private developers to install low-carbon technology.

Abbreviations: ASHP, air-source heat pump; CHP, combined heat and power; Comm., communal; DH, district heating; GSHP, ground-source heat pump; Ind., individual; kWh, Kilowatt-hour; LA, local authority; LZC, low or zero carbon; SGL, shared ground loop.

<sup>1</sup> At present the same national regulations apply to English LAs outside of Greater London (Tomaney and Colomb, 2018; GLA, 2019; MHCLG, 2021a; Greater London Authority Act, 1999).

# 2 The shift to low-carbon urban heat as a socio-technical transition

Our study understands residential heating in urban situations as an example of a socio-technical system (STS). This approach has the advantage that it highlights useful considerations when exploring LA action to support the transition to LZC heating locally. Socio-technical transitions refer to the shifting of sociotechnical systems from one state to another (Markard et al., 2012), and transition researchers recognise that technologies do not exist in isolation but rather are part of wider systems that include individuals and firms, supply chains, infrastructures, markets and regulations, norms, and traditions (Arthur, 1989; Rip and Kemp, 1998; Geels, 2002, 2010). The STS for UK residential heat supply includes technologies such as gas boilers and the extensive national gas grid, novel technologies such as SGLs, heat pumps and heat networks, national and local policies and regulations, users with existing practises and preferences, and heating installers who work with established networks (Chaudry et al., 2015; Barnes and Bhagavathy, 2020; Simpson et al., 2020; Barns et al., 2021). Sociotechnical transition studies have tended to focus on a technologyled transition and assume a national scale (Coenen et al., 2012), although there are exceptions that we discuss later. This has led to a lack of attention to place-based factors that shape local transitions. Cities have been identified by some scholars as important sites where local policy interventions can impact the deployment of low-carbon heat technologies, such as heat networks (Sullivan et al., 2013; Bush, 2016; Webb et al., 2016). For these reasons, we explore socio-technical factors and their impact on action by LAs through the use of frameworks from the socio-technical transitions literature.

Frameworks such as the multi-level perspective (MLP) have been developed over several decades to help those seeking to understand socio-technical transitions. The MLP considers sociotechnical transitions primarily through the effect of interactions of three levels of socio-technical systems-niche, regime, and landscape (Geels, 2002; Geels and Schot, 2007). Stability emanates at the regime level through shared norms, rules, beliefs and expectations, which guide the behaviour of actors and can lead to a lock-in of dominant technologies and infrastructures. The stability and high carbon lock-in of UK residential heating are partly due to the regime characteristics highlighted in Section 1 (Taylor et al., 2013). Innovations such as LZC heating can develop in niches that are shielded from the pressures of the incumbent regime. LZC niche support has been provided by public funding streams such as the UK's previous domestic (2014-2021) and nondomestic (2011-2021) Renewable Heat Incentive and the current Boiler Upgrade Scheme (Ofgem, 2021b,c, 2022; BEIS, 2022). These schemes reflect attempts by the UK government to support niche innovation heating technologies by incentivising consumer uptake to compete against incumbent fossil-based options (Lowes et al., 2019; Martiskainen et al., 2021; BEIS, 2022). The socio-political regime concept has been expanded to help consider the power relationships that incumbent regimes may use to resist change (Geels, 2014; Swilling et al., 2016; Feola, 2020). Research on incumbency in the UK energy sector has identified a vigorous coalition of actors at a national level, primarily from the gas industry and established supply companies, working to influence the discourse towards the perceived benefits of retaining the advantages of gaseous fuel through a move to hydrogen (Bolton and Foxon, 2011; Lowes et al., 2019, 2020b; Lockwood et al., 2020). Whilst studies have also tended to focus on a national scale, they suggest value in considering incumbent action and existing power relationships that may resist change at a local level.

Several scholars have sought to apply a socio-technical transition analysis to the city scale (Hodson and Marvin, 2010; Verbong et al., 2010; Raven et al., 2012; Kivimaa, 2014; Truffer et al., 2015; Bush et al., 2017). A general theme across this literature is the crucial role of city actors in creating and nurturing local niches where innovations can develop. For example, through a study of urban district heating as a niche innovation, Bush et al. (2017) found that LAs were key to enabling greater deployment through setting strategic and spatial planning policies that require district heating consideration, as well as by supporting teams responsible for connecting and persuading local stakeholders internally and externally, building social networks required to deliver projects, etc. An analysis of UK LAs found that divergent abilities between large unitary urban authorities and lower-tier district boroughs were more common in rural settings (Tingey and Webb, 2020). A place-based socio-technical analysis of Bristol since 1966 found a strong history of local grassroots activism, which has supported municipal initiatives on the environment (Torrens et al., 2018). This contributed to the establishment of the Bristol City Council's Sustainable Cities Team in 1994 to coordinate sustainability-related activities from different departments and develop a strategic direction. Schwanen (2015) provided a useful template for this work through a comparative case study of the socio-technical transition in low-carbon mobility in two other UK cities, Oxford and Brighton. Schwanen (2015) found that although city contexts were shaped and limited by the national regime, local LA action was possible, shaped by local political and historical contexts, e.g., the importance of local politicians in driving forward specific innovations.

Overall, we propose that LZC heating technologies can be usefully considered as niche innovations subject to a challenging socio-political regime exhibiting a lock-in of high-carbon fossilbased heating. This can help explain low levels of deployment in the UK. Whilst this is a national rather than local or cityspecific challenge, there is increasing recognition of the role of cities and city actors in creating and supporting local niches where the conditions for deployment can be fostered. We found the examination of these national-local dynamics employing the concepts and principles of socio-technical transitions to be particularly helpful in our analysis. For example, as outlined in Section 4, this helped to identify that in acting as local incumbents, developers sought to invoke the national regime to challenge LA attempts to implement planning system measures to support LZC niche innovations.

### 3 Materials and methods

We investigated how LAs were able to use the planning system to encourage developers to choose LZC heating. To do this, we adopted an embedded case study comprising 30 residential

Bristol			Leeds		
Identifier	Outset technology	Final technology	Identifier	Outset	Final technology
ASHP1	ASHP	ASHP	CITYDH3	Heat network	Heat network
ASHP2	Gas boiler	ASHP	DIRELEC2	Gas communal	Panel heaters
ASHP3	Panel heaters	ASHP	DIRELEC3	Panel heaters	Panel heaters
CITYDH1	Heat network	Heat network	DIRELEC4	Panel heaters	Panel heaters
CITYDH2	Gas boiler	Heat network	DIRELEC5	Panel heaters	Panel heaters
DIRELEC1	Panel heaters	Refused	DIRELEC6	Panel heaters	Panel heaters
GASCOMM1	Gas communal	Gas communal	DIRELEC7	Unknown	Panel heaters
GASCOMM2	Gas communal	Gas communal	DIRELEC8	Panel heaters	Panel heaters
INDGAS1	Gas boiler	Gas boiler	DIRELEC9	Panel heaters	Panel heaters
SGL1	Gas boiler	SGL	GASCOMM4	Gas communal	Gas communal
SGL2	ASHP	SGL	INDGAS2	Gas boiler	Gas boiler
SGL3	SGL	SGL	INDGAS3	Gas boiler	Gas boiler
SGL4	Biomass communal	SGL	SGL7	SGL	SGL
SGL5	ASHP	SHL	SGL8	SGL	SGL
SGL6	SGL	Refused	UNKNOWN1	Unknown	Unknown

TABLE 1 Summary of developments by case city, identifier, and heating technology.

developments across two UK cities underpinned by the sociotechnical transitions literature established in Section 2.

Leeds and Bristol were selected as the two case cities because they are both:

- Major UK cities with large populations-465,866 in Bristol and 798,796 in Leeds (Bristol City Council, 2020; Leeds City Council, 2020; ONS, 2021).
- Single-tier unitary authorities of the type recognised to be more likely to be "leaders" in energy engagement (Tingey and Webb, 2020).
- Subject to the same regulatory framework which covers all of England outside London (Greater London Authority Act, 1999; Tomaney and Colomb, 2018; GLA, 2019; MHCLG, 2021a).
- Similarly publicly committed to achieving net-zero carbon emissions by 2030 (Bristol City Council, 2018; Leeds City Council, 2019).

Bristol was chosen as the critical case due to evidence in prior work suggesting successful local policies and practises were enabling the deployment of LZC heat technologies (Torrens et al., 2018; Barns, 2022). Leeds was chosen as a comparator case with evidence suggesting good progress in active investment and implementation, but to date, there is little evidence of the local authority using the powers of the planning system to compel developers to select low-carbon heating options. A case study approach was chosen to investigate in-depth place-based contexts since it was considered likely to be key to developing an understanding of why deployment is happening in some cities and not others (Yin, 2018). To avoid researcher bias and allow each case city to be considered on its own terms, we adopted a replication logic approach. This meant each case was analysed in isolation before results were brought together to compare through a cross-case synthesis.

### 3.1 Data collection

Between June 2020 and May 2021, we collected data by building a repository of documentary evidence on 30 residential developments, with 15 in each case city, as well as relevant local strategic and policy documents. The case boundary was set as the LA geographical limit in each city. We limited the search to residential developments comprising over ten dwellings as this tends to be the lower limit for planning the application of carbon reduction policy measures. A "typical case" approach to purposeful sampling was applied to reflect the situation in each city (Emmel, 2013). In this way, we attempted to include a range of residential developments that broadly reflected the type of projects and technology choices progressing through the planning system in the case cities rather than being unusual or which could be considered outliers. We actively chose cases to include in the sample rather than allowing this to be entirely random so that we could capture a range of cases, particularly where initial sampling indicated there were interactions between LA planning officers and developers. Therefore, the approach can be considered a purposeful sampling strategy. Table 1 shows a summary of projects included in the sample.

In addition to providing an identifier for each residential development, Table 1 includes the heat delivery technology at the outset and final (at the time of data collection) stage in the planning process. This forms the basis of Section 4.3 and is important for the analysis because it highlights where the developer technology choice changed during the process. We gathered information on

TABLE 2 Interviewee identifier and role description	on.
---	-----

Identifier	Interviewee description		
DEV-1	Senior Development Manager for local authority		
DEV-2	Regional Development Manager for a large construction company delivering housing developments		
CON-1	Energy and Sustainability Specialist for energy consultant responsible for producing energy strategies		
PLANNING-1	Senior Planning Policy Officer with responsibility for climate-related planning policy		
CON-2	Principal Sustainability Consultant for energy consultant responsible for producing energy strategies		
DEV-3	Mechanical, Electrical, and Heating Manager for local authority responsible for heating and retrofit		
DEV-4	Regeneration Team Leader for community development trust		
BUILD-1	Associate Director for a medium-sized construction company		
CON-3	Director of energy consultant responsible for producing energy strategies		
DEV-5	Managing Director of private developer based on a low-car, low-energy, high-density design		
PLANNING-2	Principal Planner with responsibility for climate-related planning policy		
DEV-6	Founder and current resident of community-led low-energy housing development		

the case study cities and housing developments through a desk survey of documentary data from each city's planning portal, published minutes and recordings of local authority meetings, with a summary included in Supplementary material to this article. We also carried out expert interviews with stakeholders representing local authority planning teams, housing developers, building firms, and energy consultants. The process followed recruitment, consent, and data management practices subject to University of Leeds ethics guidelines and approval (ref LTSCPE-004) to carry out 12 interviews. A summary of interviewees is shown in Table 2.

The direct quotes in Sections 4, 5 are identified as a development (as per the identifier in Table 1) to a case city (LA1 and LA2) or interviewee (as per the identifier in Table 2). As per the terms of our ethical approval, we used identifiers to mask identities rather than directly linking subcases to named projects or cities. We also numbered each document within its reference to a case, such that CITYDH2-4, for example, refers to qualitative data source #4 in the sample related to the CITYDH2 project. A summary of documentary reference material is provided in the Supplementary material.

### 3.2 Data analysis

We examined documentary and interview evidence to construct a detailed picture of each subcase, including the type of heat technology initially specified and whether this changed during the development process. The attribute framework we developed for this analysis is included in Supplementary material. Data analysis involved both quantitative and qualitative investigation. A quantitative analysis was applied to assess the carbon reduction impacts of local planning policy and practice. A time-series analysis following Lee (2012) enabled to establish the sequence of events and interactions between developers, energy consultants, and local authority officers during the development process. We were particularly interested in exploring if the heating technology specified for the development changed during the process from individual gas boilers to LZC alternatives. A more focused exploration of the documentary and interview evidence around the time of any change in the decision was undertaken to search for causal factors in policy and practice.

To further explore causal factors behind developer technology decisions, we applied a template approach to the thematic analysis of documentary and interview evidence (King, 2017). Here, we employed a socio-technical transitions framework to form the basis of an initial coding template for application to a subset of data. In the initial coding template, we used a priori themes from the STS literature discussed in Section 2 to help identify factors shaping the deployment of LZC heating technologies at a local level. This included, for example, how developers as local regime actors were responding to the national regime for residential heat. We developed and evolved the coding template as we applied it to more data, clustering and adding themes as we identified them in the data, and applying them within a logical STS hierarchy. Ninety documents were included in the analysis-45 from each case city, including the interview transcripts; documents were added until there were no additional insights to be gained. This analysis generated evidence on the causal factors driving developer decisions on technology selection and allowed us to draw out the insights as presented in Sections 4, 5.

# 4 Results

The analysis of 30 residential developments across the two case study cities revealed that whilst shaped and limited by the national socio-political regime, local authorities can still design and enact local policy to support the transition to LZC heat technologies through niche empowerment activities. Beginning with an analysis of the broader socio-political regime in which LAs operate in Section 4.1, Sections 4.2 to 4.5 explore how LA actions in the two cities explain developer technology choices and what lies behind the changes from fossil to LZC heating, where this occurs.

# 4.1 National context and the impacts of the socio-political regime

Our analysis of the interviews and documentary evidence revealed that the national socio-political regime shapes LA action at the local level. This has, to some extent, limited LA attempts to impose more rigorous carbon reduction standards as we describe here, but importantly, it allowed LAs to implement technology-specific measures such as preventing the installation of gas boilers. There was prominent evidence to suggest LAs were constrained by national government policy that prevented them from going beyond a carbon reduction requirement of 20% better than current building regulations (MHCLG, 2015). As explored in Section 4.3.1, we found this limit impacted heat technology deployment as it can be met by the addition of solar PV without the need to adopt LZC heating. The impact was exacerbated by the unexpected nature of the policy shift to abolish the zero-carbon homes standard in 2015 through a ministerial statement (MHCLG, 2015) and uncertainty over whether local authorities had any right to demand carbon reductions beyond those stipulated in building regulations [PLANNING-1, PLANNING-2]. This caused additional challenges and delays which held back the implementation of new local plans.

With a less direct policy-related effect, planning officers highlighted the conflict they faced between a desire to implement higher standards and the risk that higher associated costs for developers might cause them to focus on other locations with less ambitious standards [PLANNING-2]. Aside from the local economic consequences, there was concern that higher standards could lead to lower housing development numbers, and potentially missed national targets resulting in a process known as "presumption in favour of sustainable development" (Upton, 2019). This is a technical status within the national English planning framework that means LAs must generally grant planning permission unless the land is protected (MHCLG, 2021a), in effect tilting the balance further in favour of development and potentially away from higher standards. In addition, LAs who challenge developers have to defend their decisions through a legal process if the developer appeals to the Secretary of State, who typically appoints a Planning Inspector to hear the case (Rankl, 2024). This can result in the LA potentially facing large legal bills as planning inspectors have the power to award costs against either party if they believe they have behaved unreasonably (Rankl, 2024), [PLANNING-1]. This has contributed to risk aversion when advocating for higher standards. In addition to the tilted balance of the planning system, an additional dynamic was the process of land valuation calculations which was believed to drive standards lower in areas with comparatively lower land values [DEV-2]. This is because when appraising the amount they may be willing to pay for a site, developers typically follow an income minus costs approach known as gross development value or GDV (Crosby and Wyatt, 2019). This includes income from the sale or rental of the homes, minus land costs, construction costs, and a profit margin, to reach a site appraisal value. This can mean in areas with lower land values and lower expected income, the additional cost of installing measures such as LZC heating would impact the GDV to a greater extent, making a location requiring this less attractive. Evidence at the time of the study found Leeds had a 33% lower average land value than Bristol (£2,150,000/ha in Leeds compared to £3,250,000/ha in Bristol) (MHCLG, 2020). At any given site, the GDV approach can also have the effect of driving competing developers to seek the lowest possible construction costs or risk losing the land and opportunity to build the development to a competitor who builds to lower standards and can, therefore, offer a higher price for the site [DEV-5].

Policy detail	Bristol	Leeds
Policy name	BCS14	EN1
Carbon reduction requirement	20% reduction in residual emissions after energy efficiency measures	20% reduction in emissions over Part L
Policy applicable to	All developments	Residential, over ten dwellings
Carbon reduction baseline measure	Building Regulations Part L 2013	Building Regulations Part L 2013
Eligible technologies	Range including non-heating, e.g., solar PV	Range including non-heating, e.g., solar PV
Fabric efficiency requirement	Yes, but no % target	Yes, but no % target
Energy generation requirement	Yes, but no % target	Yes, 10% of onsite needs

# TABLE 3 Summary of carbon reduction planning policies in Bristol and Leeds.

### 4.2 LA policies in each location

Our analysis of local policy design found that local policies were very similar in both case cities, shaped by the socio-political regime dynamics noted in Section 4.1. Both Bristol and Leeds featured planning policies that sought to reduce carbon emissions and specify both eligible and ineligible heating technologies for planning approval.

#### 4.2.1 Carbon reduction policies

Both LAs had adopted a policy requiring developers to demonstrate how their scheme would deliver a 20% reduction in carbon emissions against the national building regulation baseline.<sup>2</sup> In Bristol, this was first included in the planning policy in 2011 (Bristol City Council, 2011), whilst, in Leeds, it was included as Supplementary Planning Guidance from 2011 onwards before becoming a planning policy in 2014 (LCC, 2011, 2019). Table 3 summarises key aspects of the carbon reduction policies in both cities.

Policies were similar in both locations, with both implementing up to the maximum allowed 20% carbon reduction. Importantly, for the results set out in Sections 4.3–4.4, both policies were tied to Building Regulations 2013 as a baseline measurement for a notional building. The impact of this is explored in Sections 4.3–4.5.

#### 4.2.2 Heating technology policies

Both cities also featured heating hierarchy requirements that were implemented into policy at the same time, prioritising certain heating technologies over others, especially heat networks, and excluding gas boilers and direct electric resistive heating. Table 4 provides a summary of the equivalent policies in each location.

<sup>2</sup> At the time of the research, the baseline was set by Building Regulations Part L 2013.

TABLE 4	Summary of heat technology planning policies in Bristol and
Leeds.	

Policy detail	Bristol	Leeds
Policy name	BCS14	EN4
Headline policy requirement	Heat hierarchy of technology options	Heat hierarchy of technology options
Policy applicable to	All developments	Where technically viable, appropriate for development, in an area with sufficient existing or potential heat density
Eligible technologies	<ul> <li>Connection to district heat network</li> <li>Site-wide heat network</li> <li>Shared ground loops</li> <li>Individual building renewable heating</li> </ul>	<ul> <li>Connection to district heat network</li> <li>Site-wide heat network</li> <li>Collaboration with neighbouring sites</li> <li>Develop a district heating network</li> </ul>
Ineligible technologies	- Individual gas boilers - Direct electric panel heaters	- Individual gas boilers - Direct electric panel heaters

Although the heat hierarchy policies were broadly similar, it can be seen in Table 4 that in Bristol, the policy applied to all new developments, whilst in Leeds, the policy was prefaced with: "Where technically viable, appropriate for the development, and in areas with sufficient existing or potential heat density" [LA2-1], with a requirement for developers to submit assessments along with planning applications. Both policies offered a list of options intended to support the deployment of heat networks and did not include gas boilers or direct electric resistive heating.

Whilst not formally included in the policy hierarchy, we included SGLs in the list of eligible technologies in Bristol because documentary evidence suggests LA support for the approach in the language in LA-developer exchanges, such as: "... micro-district approaches [...] meets the definition of site-wide renewable heating within the heat hierarchy and are therefore supported by policy BCS14" [SGL4-5].

The ineligibility of gas boilers in the heat technology policies was justified citing their significant carbon emissions impact. Similarly, direct electric resistive heating was made ineligible because they it was relatively inefficient compared to heat pumps, caused a strain on the electricity grid, and ruled out future connections to district heat networks [PLANNING1, LA1-2].

# 4.3 Outcomes of LA planning policy on developer decisions

We analysed development projects for evidence to determine if LA policy and practice had an effect on developer decisions. This included mapping specified heat technologies at the initial submission, during the planning process, and at the final submission. Figures 1, 2 show the heat technologies specified in both cities at the outset and final stages of the development process. These illustrations are based on data summarised in Table 1 but shown here as Sankey diagrams to illustrate visually how the two locations differed.

TABLE 5	Compliance and impacts of carbon reduction policy	in Bristol
and Leed	ds.	

Compliance detail	Bristol (14 applicable subcases in sample)	Leeds (12 applicable subcases in sample)
20% carbon reduction policy met–initial submission	14	6
20% carbon reduction policy met–final/post- intervention submission	14	9
Max. carbon reduction–(initial/final or post-intervention)	-64%/-83%	-48%/-58%
Min. carbon reduction – (initial/final or post-intervention)	-20%/-20%	-12%/-14%
Av. carbon reduction – (initial/final or post-intervention)	-34%/-40%	26%/-29%
Technology used to achieve carbon reduction	Heating (5), solar PV (8)	Heating (6), solar PV (5)

Figures 1, 2 suggest that developer decisions in Bristol changed more frequently during the development process compared to Leeds, which had more stable conditions. The Leeds situation is also characterised by the specification of direct electric panel heaters, whilst in Bristol, there is a mix of different approaches and a trend away from gas boilers to LZC technologies, particularly SGLs. Planning submissions and records of interactions between developers or their energy consultants and LA planning officers were analysed to assess how the policies impacted developer decisions.

#### 4.3.1 Outcomes of carbon reduction policy

To first assess the impact of carbon reduction policies on technology choice, Table 5 summarises the number of subcases that met the requirement and whether the carbon savings resulting from the proposed heat technology were used by the developer to comply with the policy.

Table 5 shows that all developments met the carbon reduction requirement in Bristol, both at the initial submission and final submission stage. A lower level of compliance was found in Leeds, with 9 of the 12 applicable developments meeting the policy at the final stage. This marked an increase from the initial submission stage, suggesting measures were in place to address policy non-compliance, a result discussed further in Section 4.5. Higher maximum and average carbon reduction levels were demonstrated in Bristol, suggesting overall greater success in policy implementation.

To assess the extent to which carbon reduction requirements impacted developer heating technology decisions, we examined energy strategy documents to ascertain whether the emissions savings from the heating technology (compared to the building reductions baseline) counted towards meeting the carbon target.



Developer technology decisions in Bristol from the outset to the final stage.



Five developments in Bristol and six in Leeds used the specified heat technology to meet the carbon reduction target. However, 13 other schemes did not reference heating to satisfy carbon reduction requirements. This suggests that in both locations, the carbon reduction requirements of 20% (the maximum allowed under the current regime) can be met through other measures such as the addition of solar PV. As a result, the carbon reduction policies were not impactful in encouraging developers to choose LZC heat technologies.

#### 4.3.2 Outcomes of heat technology policy

We examined the impact of the heat technology policies implemented in both locations to ascertain whether meeting the heat hierarchy policy played a role in developer technology choice. Table 6 shows a summary of heating technologies specified in the developments and whether they met the local heating policy.

Table 6 illustrates that the heating technology policy was aligned with developers opting for LZC heating technologies in Bristol but less so in Leeds. All but one development in Bristol

TABLE 6 Heating technologies specified in Bristol and Leeds with heating policy compliance.

Policy approved technology	Bristol	Leeds
Shared ground loop	6	2
ASHP	3	0
Connection to district heat network	2	1
Site-wide network (non-renewable)	2	1
Total	13	4
Total where policy applied	13	2
Policy non-approved technology		
Gas boilers	1	2
Direct electric panel heaters	0	8
Unknown	0	1
Total	1	11
Total where policy applied	1	11

met the heating policy requirement through the technologies shown in Table 6, with six developments specifying SGLs to achieve BSC14 compliance (although one was refused planning permission for other reasons). One development achieved planning permission without meeting the policy requirement (INDGAS1), with individual gas boilers specified and the application approved at initial submission with no evidence of challenge. Eight projects initially proposed non-compliant technologies, such as gas boilers and direct electric resistive heaters, prior to final compliant specifications and planning approvals. This suggests that in Bristol, the heating policy was successfully implemented to shape developer decisions towards LZC heating when combined with active intervention measures (explored further in Section 4.5).

In Leeds, two developments fully met the heat policy through connection to a future district heat network and a site-wide communal heat network served by gas boilers. However, in six developments, no reference to the heating policy or evidence of compliance was evident, and it was not clear from the submitted materials whether the development was considered appropriate for policy applicability. These schemes primarily specified direct electric resistive heating. The evidence suggests a less successful implementation of a heating policy in Leeds.

# 4.4 Developer attempts to resist LA planning policy

An important factor in LZC technology deployment was how developers responded to the policy design and implementation approach of the LAs under investigation. We found developer decisions shaped by the national and local regimes, with evidence that developers attempted to invoke stabilising regime dynamics and resist LA efforts to encourage them to choose LZC heating options.

Our analysis found that developers frequently used carbon factors inherent within national building regulations to challenge local policy in both cities. The electricity system in the UK is rapidly decarbonising due to the introduction of renewable technologies, but building regulations have struggled to keep pace with these developments (Rees, 2019). At the time of the study, the in-force National Building Regulations Part L 2013 (with inherent SAP2012 methodology) featured electricity grid carbon emissions per kWh approximately two times that of natural gas per equivalent unit, with the to-be-introduced Part L 2021 Uplift (with inherent SAP10) effectively reversing this situation (MHCLG, 2018; DLUHC, 2022). Table 7 summarises the instances identified against the two versions of SAP and how the carbon factors should justify diverging from local policy requirements.

Table 7 shows this type of incumbent challenge to be a more prominent feature in Bristol, likely in response to the more active intervention measures explored in Section 4.5. Importantly, whilst older national building regulations were used to justify gas boilers because of the apparent lower carbon content of natural gas compared to electricity, newer national building regulations with a lower grid carbon factor were used to justify the specification of direct electric panel heaters.

Aside from the direct challenge of planning policy, there was a clear trend for developers to increasingly opt for direct electric panel heaters. These were proposed in ten projects in the sample. Interview evidence from energy consultants who work with developers to propose heating technologies for their schemes described how this was primarily because developers favour the most straightforward and lowest cost technology to source and install: "[developers] want easiest and cheapest, they always want cheapest!" [CON-3]. It was understood that previously this was gas boilers, but increasingly this will be electric panel heaters in the future as national building regulations are updated and there is growing recognition of the need to move away from gas: "it's just a panel on the wall, they don't want to bother with a wet system in the apartments, so they want electric panels, and they think that's going to be easy and cheap and they, well cheap basically!" [CON-3].

There was also evidence that developers attempted to resist change locally by appealing to elected councillors directly. As part of the normal governance arrangements for local authorities, a few locally elected councillors serve as members of a planning committee. These committees decide on planning applications with advice and guidance from planning officers (Rankl, 2023). In Leeds, currently, there are three planning committees-the North and East Plans Panel, the South and West Plans Panel, and the City Plans Panel (LCC, 2023). In Bristol, there are two planning committees-Development Control A and B (BCC, 2023). Whilst most planning decisions are made by planning officers through delegated powers, larger and more complex planning applications of the type included in the study are decided through these committees. Planning committee members take into account the detailed officer's reports about each proposal along with their recommendations to approve or refuse applications. However, councillors are not bound to the recommendations made by the officers and can use their own judgement. This opens up potential opportunities for developers to influence the councillors while they

Building regulations	Methodology and carbon factors	Bristol, no.	Leeds, no.	Justification
Part L 2013	SAP 2012, electricity 0.519 kgCO <sub>2</sub> /kWh, natural gas 0.210 kgCO <sub>2</sub> /kWh	6	0	Lower carbon factors in natural gas mean gas boilers should be allowed
Part L 2021 Uplift	SAP 10, electricity 0.136 kgCO <sub>2</sub> /kWh, natural gas 0.216 kgCO <sub>2</sub> /kWh	3	2	Lower carbon factors in grid electricity mean direct electric panel heaters should be allowed

TABLE 7 Number of subcases where developers used carbon factors to challenge local planning policy.

make the decisions. For example, a developer interviewee described how they had tried but failed to persuade a local councillor to allow them to implement a non-compliant heat solution despite being challenged by the officer. The councillor had responded: "that ain't changing [...] you're going to have to find another solution" [DEV-4].

Overall, the findings indicated that when faced with policies encouraging LZC heating technologies, developers attempted to challenge these to specify non-compliant heat technologies. This included invoking elements of the national regime, such as carbon factors inherent in building regulations, as well as lobbying councillors in their role as planning committee members and making planning decisions. Given these conditions, Section 4.5 explores how Bristol was able to encourage developers to choose LZC heating technologies.

#### 4.5 How the results were derived

It was established in Sections 4.1–4.4 that both locations were subject to the same regime impacts, had broadly similar planning policies in place, and experienced developers attempting to challenge policy measures to choose LZC heating technologies. However, the evidence suggested greater success apparent in Bristol in driving developers to choose LZC heating. We further analysed the case studies to look for evidence which could explain the difference and suggest how Bristol achieved better policy outcomes. This included examining the interactions between planning officers and developers for instances of direct engagement and intervention in the planning process.

Table 8 shows a summary of identified LA intervention activities and the instances where this supported a change in heating technology. In the table, "direct intervention" refers to activities where direct contact between planning officers and developers was identified. This included records of emails and evidence of meetings having taken place. "Indirect intervention" refers to instances where measures operated through the planning process, such as placing conditions on planning approvals for the developer to provide more information about how they would meet the policy prior to construction beginning.

The results in Table 8 show greater evidence of direct intervention in Bristol as well as associated changes in heating technology. In Bristol, identified action was undertaken by a multidisciplinary team of officers who were external to the planning function and acted in an advisory capacity as well as a statutory consultee on planning applications, much in the same way as the fire service or water authority. Evidence showed that the Sustainable Cities Team had the power to demand changes to the specified heat technology and to recommend the refusal of planning permission for failure to meet policy requirements. An example of one such intervention is illustrated in a communication from officers in response to continued attempts by the developer to opt for direct electric panel heaters, resulting in planning refusal on these grounds:

"We [the Sustainable Cities Team] remain strongly of the view that our implementation of the heat hierarchy, which excludes electric resistive heating, aligns with government policy and thinking on the decarbonisation of heat, and the independent advice we have received on the decarbonisation of heat in Bristol...For these reasons, we continue to object to this application" [DIRELEC2-5].

Other more supportive interventions by the Sustainable Cities Team took the form of engaging with the developers to explain the heating policy requirements, recommending certain LZC options as a suitable option for some developments, and offering financial support for operations and maintenance costs if they were to choose the preferred LZC option [SGL2-9, SGL2-10]. Technology recommendations by officers were identified in seven subcases and included SGLs, ASHPs, and connecting to a district heat network. Overall, policy non-compliant technologies were changed to compliant options due to active intervention in six instances. Interventions were identified in 11 instances and classified into three broad types (with numbers of each in brackets):

- 1. Advising planning refusal for failure to meet policy (3).
- 2. Compliance activities to ensure technology policy was followed (5).
- 3. Guidance and support for how developers could meet heating policy (3).

Evidence from Bristol showed that officers were supported by the local political leadership to engage in these intermediary activities in cases where developers attempted to circumvent or challenge their authority [DEV-4]. Overall, we found that Bristol's Sustainable Cities Team was a key factor in the relative success of the city in encouraging developers to choose LZC heating technologies.

In Leeds, ten interventions by planning officers were identified across the sample, with two identified as direct interventions similar to that observed in the Bristol sample. In the DIRELEC4 development, for example, a report from officers to the planning committee stated: "Through discussions and negotiations with

Case city	Direct intervention	Indirect intervention	Change in technology	No change identified
Bristol	10	1	9 (3 refused)	1
Leeds	2	8	3	7

#### TABLE 8 Summary of LA intervention and outcome in Bristol and Leeds.

the developer, the scheme is now considered to be compliant with Policy" [DIRELEC4-6]. This suggests that planning officers were engaging with applicants to try to encourage them to meet policy requirements, but Leeds did not benefit from an equivalent statutory consultee approach from a team with the resources and technical expertise to challenge developer decisions.

In eight other Leeds subcases, evidence from planning documents suggested some intervention via an indirect process. The intervention took place indirectly in the form of a report from the planning officers to the planning committee (which included elected local city councillors making the decision) recommending certain conditions placed on the award to further clarify how policy requirements would be met. Three subcases were identified where this approach resulted in a change or clarification of technology. This suggests limited success from this technology assessment and evaluation activity. However, there was no evidence of interventions leading to a change in heat technology. Three developments achieved local planning permission despite incomplete or incorrect information. There was some evidence that the situation was beginning to change in Leeds with more active intervention since the LA's declaration of climate emergency in 2019: "... in the light of the Climate Emergency declaration in March 2019, the minimum figures as adopted by Full Council in the Core Strategy were insisted upon." [DIRELEC4-6].

Figure 3 shows a summary of the developer journeys in both case cities. Whilst both locations featured similar policy inputs to the process, it led to different outcomes through more active engagement and intervention in the planning process. In Bristol, heat technology planning policies were combined with action from the Sustainable Cities Team and support from elected politicians to create the conditions whereby developers were more likely to choose LZC heating options, thus creating the conditions for local heat transition.

Summarising the results across Sections 4.1–4.5, we found that the LA in Bristol had put in place four conditions that contributed to the successful deployment of LZC heat technologies:

- 1. Implemented a heat technology planning policy that excluded conventional technologies of gas boilers and direct electric heating.
- 2. Empowered local authority officers to enforce the heat technology planning policy.
- 3. Supported developer decisions through awareness raising of LZC options and providing advice and support to consider them.
- 4. Provided political commitment to challenge developers who sought to invoke regime dynamics to resist local planning requirements.

The evidence we found indicates that implementing more of these elements increases the likelihood of deploying LZC heat technology. When only one element was in place, for example, a heat hierarchy policy in Leeds, but was not supported by other intermediary activities, the outcome was that conventional heat technologies were more likely to be installed.

# **5** Discussion

This study set out to explore what actions LAs can take to shape the deployment of LZC heat technologies in residential settings and the effect this has. Overall, the study found that local authorities were able to implement policy and practice measures within the current regulatory framework, which helped bring forward LZC heating options. These were being employed successfully in Bristol, and developers were encouraged to choose LZC heating for new developments.

A key measure within the planning system was a heating technology policy specifying a certain range of heating technology options and, importantly, prohibiting conventional gas boilers or direct electric heating. This was effective in driving developers to choose LZC heating technologies when combined with active intervention, support, and enforcement. Our findings suggest the both the design and implementation of local policies are subject to a broader socio-political regime, which impacts the LA's ability to take such steps. In addition, developer technology decisions and how developers respond to local policy and practise are itself shaped by wider factors. The English planning system enabled LAs to take measures that encouraged the uptake of LZC, but at the same time, it also placed distinct limits on them. For instance, LAs are required to limit carbon reduction to no more than 20%.

There was evidence that residential developers generally favoured the lowest cost and lowest hassle heat technologies. When challenged by local LA planning policies to encourage the adoption of LZC heating, developers attempted to resist by leveraging the institutional framework of National Building Regulations. Under prior national building regulations, higher carbon factors for electricity compared to natural gas tended to be applied to individual gas boilers. However, evidence shows developers are increasingly likely to opt for direct electric panel heaters and use the lower carbon factors of grid electricity as justification to challenge local attempts to force them to choose LZC alternatives.

The trend towards direct electric panel heaters may increase as national policy recognises the reducing carbon content of grid electricity (DLUHC, 2023). With the much higher power consumption compared to LZC alternatives, this carries challenges and risks for the electricity grid, especially when combined with the electrification of other sectors, including transport (Lowes et al., 2020a; Chaudry et al., 2022). Electric resistive heating is not generally featured in UK heat decarbonisation scenarios, and an unplanned technology shift towards this technology may



pose risks to the UK energy system and decarbonisation efforts (Slorach and Stamford, 2021). Whilst the details of the new building regulations—the Future Homes and Buildings Standards—are yet to be confirmed, this study's evidence suggests that local planning authorities keen to prioritise LZC heating options in new developments can successfully implement local policies and practises within the current national regulatory framework that will enable them to do this.

The study found evidence that the English planning system places considerable power in the hands of developers and tilts the balance away from LAs and their ability to demand higher standards. This has been supported by central government housing delivery targets for LAs, which effectively favour volume housebuilders. The developer valuation and appraisal approach that favours the lowest cost and equivalent lowest standards and penalises those developers keen to take a more progressive approach also potentially makes it more challenging for local authorities with lower land values to implement higher standards. Overall, our study emphasises the challenges faced by planning authority officers to simultaneously design, implement, and enforce environmental building standards in support of organisational commitments whilst also promoting local economic development and growth.

Our findings also suggest that Bristol's Sustainable Cities Team was a key factor in the relative success of the city in encouraging developers to choose LZC heating options through a dual role of supportive and enforcement activities. It is important for wider application to consider whether the conditions put in place in Bristol could be replicated elsewhere. Torrens et al. (2018) described how over four decades Bristol's approach developed through co-evolution between urban experimentation and governance, leading to the establishment of the multidisciplinary Sustainable Cities Team in the 1990s to work across local authority departments and intermediate between the city's environmental groups and the council. The existence of the Sustainable Cities Team, the measures they were empowered to take, and the political backing needed to implement the agenda are key factors in the current success found in this study. The study's findings also indicated that comparatively higher land values in Bristol may have reduced the disincentive for developers to implement higher-cost LZC heating measures, supporting the LA's objectives to implement and enforce higher standards. Recognising that not every LA will have the distinctive history and local socio-political regime of Bristol, other LAs can create local conditions that will encourage developers to choose LZC heating options. While each condition supports niche innovations on its own, when combined, they amplify each other and work together to shape the local socio-political regime.

Since data collection ended, both Leeds and Bristol have embarked on highly ambitious programmes to update their climate and carbon-related planning policies, including further requirements around LZC heating technologies as well as embodied carbon from construction (Bristol City Council, 2022; Leeds City Council, 2022). These will require approval by the UK's Planning Inspector in line with regulatory requirements. Recent evidence finds that the Planning Inspector is willing to reject similar bold planning policies, further illustrating the challenging socio-political landscape faced by UK local authorities keen to take ambitious climate action (Ellis, 2022). With the measures set out in this study already operating successfully, there is reason to be hopeful they can be applied elsewhere in the current institutional regime. We also note that since data collection ended, Leeds City Council recruited a dedicated officer to assess climate and energy aspects of planning applications.

The study benefitted from the transparent nature of the UK's planning system, which makes all documents submitted in pursuit of planning permission publicly available. However, the challenge of gathering and analysing material limited the study to 2 case cities and 30 subcase developments. There would be a benefit from analysing other cities in the UK and beyond, as well as smaller and more rural authorities that face different regime contexts. In addition, as the measures explored in this study take place within the English (outside London) planning system, the findings are applicable primarily to the new building sector rather than the enormous challenge of retrofitting homes and buildings already constructed. The ability of LAs to usher the transformative change required to retrofit existing properties is constrained by the lack of a regulatory framework as well as acute financial difficulties (Gray and Barford, 2018). Despite these factors, the findings are significant and offer a useful framework for LAs keen on taking action to reduce the climate impacts of the residential heat sector in their area. Many features of the institutional framework which shape the findings in the study relate to the specific context of the English planning system. However, many cities around the world are struggling with residential heat decarbonisation, challenging socio-political regimes, and

incumbency in the energy and development sectors. The findings and proposed city authority measures set out in the study are, therefore, applicable internationally.

# 6 Conclusion

In this paper, we applied a socio-technical analysis to an embedded case study of two UK cities. Through an analysis of local authority policy and practice in regard to 30 residential developments, the study demonstrated how LAs could shape the local regime for residential heating and create the conditions for low-carbon heat technologies to come forward in support of the net-zero transition. LAs are identified as key actors who need to be recognised in the analysis of infrastructure transitions. However, the ability of LAs to shape the local regime for residential heat decarbonisation is limited by the national socio-political regime, which is subject to the effects of lock-in and incumbency. The study findings emphasise the importance of considering location-based aspects of transitions and the need explicitly to recognise power and politics in the socio-technical analysis of how new infrastructure systems can be brought forward.

This study took place in the UK context of decarbonising the electricity grid and expected changes to building regulations, both of which are likely to support the shift to electrified heating options. However, the study finds that without intervention, developer technology decisions are likely to trend from gas boilers towards direct electric panel heaters, and they are willing to invoke national policies to challenge local attempts to compel them to choose more efficient but less well-established technologies. Widespread deployment of direct electric panel heaters could lead to issues of grid stability, challenge future energy system flexibility, and run counter to the need to use low-carbon electricity as efficiently as possible. The findings underscore the case for local intervention in the energy system and for the UK's heat transition. Based on this study's evidence, we recommend a series of overlapping conditions that local authorities, as key city actors in the net-zero transition, can put in place as part of the planning process for residential developments. These include a heat technology policy which excludes conventional options of gas boilers and direct electric resistive heating; resourcing and empowering officers to enforce the policy; providing advice and support to developers in their technology decision processes to consider low-carbon heat technologies, and providing political commitment and support against challenge from incumbent developers. Whilst the findings set out above must be viewed in the context of the UK (and specifically English) planning and regulatory system, they can apply to other cities and locations where local commitments to decarbonise residential heating meet challenges of incumbency, centralisation, and a liberalised economy that places significant power in the hands of developers.

### Data availability statement

The datasets presented in this study can be found in online repositories. The names of the repository/repositories

and accession number(s) can be found in the article/Supplementary material.

# **Ethics statement**

The studies involving humans were approved by the University of Leeds MEEC (Faculties of Maths, Engineering & Physical Sciences) Research Ethics Committee. The studies were conducted in accordance with the local legislation and institutional requirements. The participants provided their written informed consent to participate in this study.

# Author contributions

DB: Conceptualization, Data curation, Methodology, Visualization, Writing—original draft. CB: Conceptualization, Funding acquisition, Writing—review & editing. PT: Conceptualization, Writing—review & editing. AO: Writing review & editing.

# Funding

The author(s) declare that financial support was received for the research, authorship, and/or publication of this article. This work was supported by the UK Engineering & Physical Sciences Research Council Doctoral Training Programme, grant reference 1958986.

# Acknowledgments

We would like to thank all interview participants who gave the benefit of their time, experience, and expertise to make this study possible.

# Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

# Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

# Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/frsc.2024. 1339709/full#supplementary-material

# References

Adam, K. (2018). The Role of Collaboration in Realising Local Authority Energy Objectives: An Institutional and Stakeholder Perspective. Leeds, UK: University of Leeds.

Arthur, W. B. (1989). Competing technologies, increasing returns, and lock-in by historical events. *Econ. J.* 99, 116–131. doi: 10.2307/2234208

Bale, C., Barns, D., and Turner, J. (2022). Shared ground heat exchange for the decarbonisation of heat. Available online at: https://eprints.whiterose.ac.uk/185509/ (accessed April 27, 2022).

Bale, C. S. E., Foxon, T. J., Hannon, M. J., and Gale, W. F. (2012). Strategic energy planning within local authorities in the UK: a study of the city of Leeds. *Energ. Policy* 48, 242–251. doi: 10.1016/j.enpol.2012.05.019

Barnes, J., and Bhagavathy, S. M. (2020). The economics of heat pumps and the (un)intended consequences of government policy. *Ener. Policy* 138:111198. doi: 10.1016/j.enpol.2019.111198

Barns (2022). Unlocking the potential for thermal energy storage in the UK (phd). University of Leeds.

Barns, D. G., Taylor, P. G., E., and Bale, C. S., Owen, A. (2021). Important social and technical factors shaping the prospects for thermal energy storage. *J. Ener. Stor.* 41:102877. doi: 10.1016/j.est.2021.102877

BCC (2023). Bristol City Council Constitution: Part 3 Responsibility for Functions. Bristol, UK: Bristol City Council.

BEIS (2018). Clean Growth - Transforming Heating: Overview of Current Evidence. Department for Business, Energy & Industrial Strategy.

BEIS (2021). Heat and Buildings Strategy. HM Government.

BEIS (2022). The Domestic Renewable Heat Incentive (DHRI) closure, and its successor, the Boiler Upgrade Scheme. GOV.UK. Available online at: https://www.gov.uk/government/publications/changes-to-the-renewable-heat-incentive-rhi-schemes/ closure-of-the-domestic-renewable-heat-incentive-dhri-and-its-successor-theboiler-upgrade-scheme (accessed February 15, 2024).

BEIS and Triple Point Heat Networks (2022). Green Heat Network Fund: guidance for applicants (No. Version 2.0). Department for Business, Energy and Industrial Strategy.

Boesten, S., Ivens, W., Dekker, S. C., and Eijdems, H. (2019). "5th generation district heating and cooling systems as a solution for renewable urban thermal energy supply," in Advances in Geosciences. *Presented at the European Geosciences Union General Assembly 2019 EGU Division Energy, Resources and Environment (ERE) - EGU General Assembly 2019 Vienna, Austria* (Copernicus GmbH), 129–136. doi: 10.5194/adgeo-49-129-2019

Bolton, R., and Foxon, T. J. (2011). Governing infrastructure networks for a low carbon economy: co-evolution of technologies and institutions in UK electricity distribution networks. *Compet. Regul. Netw. Industr.* 12, 2–26. doi: 10.1177/178359171101200101

Bristol City Council (2011). Bristol Development Framework Core Strategy. Bristol, UK: Bristol City Council.

Bristol City Council (2018). *Minutes of the Full Council - 13 November 2018 at 6.00 pm*. Bristol, UK: Bristol City Council.

Bristol City Council (2020). Mid-2020 Population Estimates Note. Bristol, UK: Bristol City Council.

Bristol City Council (2022). Local Plan Review. Bristol, UK: Bristol City Council.

Buffa, S., Cozzini, M., D'Antoni, M., Baratieri, M., and Fedrizzi, R. (2019). 5th generation district heating and cooling systems: a review of existing cases in Europe. *Renew. Sustain. Energy Rev.* 104, 504–522. doi: 10.1016/j.rser.2018.12.059

Bush, R. (2016). Governing low carbon socio-technical transitions – a case study of district heating in Great Britain. Leeds, UK: University of Leeds.

Bush, R. E., Bale, C. S. E., Powell, M., Gouldson, A., Taylor, P. G., and Gale, W. F. (2017). The role of intermediaries in low carbon transitions – empowering innovations to unlock district heating in the UK. *J. Cleaner Prod.* 148, 137–147. doi: 10.1016/j.jclepro.2017.01.129

Bush, R. E., Bale, C. S. E., and Taylor, P. G. (2016). Realising local government visions for developing district heating: experiences from a learning country. *Energ. Policy* 98, 84–96. doi: 10.1016/j.enpol.2016.08.013

CCC (2019). UK Housing: Fit for the Future? London, UK: Committee on Climate Change.

CCC (2020). The Sixth Carbon Budget: The UK's Path to Net Zero. London, UK: Committee on Climate Change.

Chaudry, M., Abeysekera, M., Hosseini, S. H. R., Jenkins, N., and Wu, J. (2015). Uncertainties in decarbonising heat in the UK. *Energ. Policy* 87, 623–640. doi: 10.1016/j.enpol.2015.07.019

Chaudry, M., Jayasuriya, L., Blainey, S., Lovric, M., Hall, J. W., Russell, T., et al. (2022). The implications of ambitious decarbonisation of heat and road

transport for Britain's net zero carbon energy systems. Appl. Ener. 305:117905. doi: 10.1016/j.apenergy.2021.117905

Clarke, L., Sahin-Dikmen, M., and Winch, C. (2020). Transforming vocational education and training for nearly zero-energy building. *Build. Cities* 1, 650–661. doi: 10.5334/bc.56

Coenen, L., Benneworth, P., and Truffer, B. (2012). Toward a spatial perspective on sustainability transitions. *Res. Policy, Spec. Sect. Sustain. Trans.* 41, 968–979. doi: 10.1016/j.respol.2012.02.014

Crosby, N., and Wyatt, P. (2019). Valuation of development property (RICS Professional Standard, global). London, UK: Royal Institute of Chartered Surveyors.

DLUHC (2022). Conservation of fuel and power: Approved Document L: 2021 edition. Department for Levelling Up, Housing and Communities.

DLUHC (2023). The Future Homes and Buildings Standards: 2023 consultation. GOV.UK. Available online at: https://www.gov.uk/government/consultations/the-future-homes-and-buildings-standards-2023-consultation (accessed December 14, 2023).

Dodds, P. E., and McDowall, W. (2013). The future of the UK gas network. *Energ. Policy* 60, 305–316. doi: 10.1016/j.enpol.2013.05.030

ECIU (2024). Net Zero Scorecard. Energy and Climate Intelligence Unit. Available online at: https://eciu.net/netzerotracker (accessed February 26, 2024).

Element Energy (2023). Low Carbon Heat Study: An assessment of the impact of ground and air source heat pump deployment and heating demand flexibility on the GB electricity system and households. Available online at: https://www.erm. com/contentassets/553cd40a6def42b196e32e4d70e149a1/low-carbon-heat-study--executive-summary.pdf (accessed May 31, 2023).

Ellis, H. (2022). "Local Plans and net zero objectives," in *Town and Country Planning* 220-223.

Ellis, H., and Fieth, J. (2021). The Climate Crisis: A Guide for Local Authorities on Planning for Climate Change. London, UK: Town and Country Planning Association.

Emmel, N. (2013). Sampling and Choosing Cases in Qualitative Research : A Realist Approach. Los Angeles, CA: SAGE. doi: 10.4135/9781473913882

Feola, G. (2020). Capitalism in sustainability transitions research: time for a critical turn? *Environ. Innov. Soc. Trans.* 35, 241–250. doi: 10.1016/j.eist.2019.02.005

Fudge, S., Peters, M., and Woodman, B. (2016). Local authorities as niche actors: the case of energy governance in the UK. *Environ. Innov. Soc. Trans.* 18, 1–17. doi: 10.1016/j.eist.2015.06.004

Ge, M., Friedrich, J., and Vigna, L. (2022). 5 Facts about Country and Sector GHG Emissions. Washington, DC: World Resources Institute.

Geels, F. (2014). Regime resistance against low-carbon transitions: introducing politics and power into the multi-level perspective. *Theory, Cult. Soc.* 31, 21–40. doi: 10.1177/0263276414531627

Geels, F. W. (2002). Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. *Res. Policy* 31, 1257–1274. doi: 10.1016/S0048-7333(02)00062-8

Geels, F. W. (2010). Ontologies, socio-technical transitions (to sustainability), and the multi-level perspective. *Res. Policy* 39, 495–510. doi: 10.1016/j.respol.2010.01.022

Geels, F. W., and Schot, J. (2007). Typology of sociotechnical transition pathways. Res. Policy 36, 399-417. doi: 10.1016/j.respol.2007.01.003

GLA (2019). What is the new London Plan? London, UK: GLA.

GLA and Etude (2018). Low Carbon Heat: Heat Pumps in London. London, UK: GLA.

Gray, M., and Barford, A. (2018). The depths of the cuts: the uneven geography of local government austerity. *Cambr. J. Reg. Econ. Soc.* 11, 541–563. doi: 10.1093/cjres/rsy019

Greater London Authority Act (1999). c. 29. Available online at: https://www.legislation.gov.uk/ukpga/1999/29/contents (Accessed July 8, 2020).

Gross, R., and Hanna, R. (2019). Path dependency in provision of domestic heating. *Nat. Energy* 4, 358–364. doi: 10.1038/s41560-019-0383-5

Guertler, P., Carrington, J., and Jansz, A. (2015). *The Cold Man of Europe*. London: Association for the Conservation of Energy.

Hodson, M., and Marvin, S. (2010). Can cities shape socio-technical transitions and how would we know if they were? *Res. Policy* 39, 477-485. doi: 10.1016/j.respol.2010.01.020

Howard, M., and Crook, T. (2021). Rethinking heat: A utility based approach for ground source heat pumps. A discussion paper by Regen – March 2021 Regen.

IEA (2020). Energy Technology Perspectives (2020). Paris, France: IEA.

King, N. (2017). Template analysis for business and management students, Mastering Business Research Methods. Sage, Los Angeles. doi: 10.4135/9781473983304 Kivimaa, P. (2014). Government-affiliated intermediary organisations as actors in system-level transitions. *Res. Policy* 43, 1370–1380. doi: 10.1016/j.respol.2014.02.007

Knobloch, F., Pollitt, H., Chewpreecha, U., Daioglou, V., and Mercure, J.-F. (2019). Simulating the deep decarbonisation of residential heating for limiting global warming to 1.5 °C. *Ener. Effic.* 12, 521–550. doi: 10.1007/s12053-018-9710-0

Krog, L. (2019). How municipalities act under the new paradigm for energy planning. Sustain. Cities Soc. 47:101511. doi: 10.1016/j.scs.2019.101511

LCC (2011). Building for Tomorrow Today: Sustainable Design and Construction, Supplementary Planning Document. Leeds, UK: Leeds City Council.

LCC (2019). Core Strategy (as amended by the Core Strategy Selective Review 2019). Leeds, UK: Leeds City Council. doi: 10.12968/cypn.2019.3.54

LCC (2021). Thousands of Leeds tenants to enjoy cheaper energy bills as council appoints contractor to deliver £24m heating upgrades. Leeds City Council News. Available online at: https://news.leeds.gov.uk/news/thousands-of-leeds-tenants-to-enjoy-cheaper-energy-bills-as-council-appoints-contractor-to-deliver-gbp-24m-heating-upgrades (accessed May 16, 2022).

LCC (2023). *Plans panel meetings*. Available online at: https://www.leeds.gov. uk:443/planning/planning-permission/plans-panel-meetings (accessed February 26, 2024).

Lee, B. (2012). "Using documents in organizational research," in *Qualitative Organizational Research: Core Methods and Current Challenges*, eds G. Symon and C. Cassell (London; Los Angeles, CA: SAGE Publications).

Leeds City Council (2019). Proceedings of the Meeting of the Leeds City Council held Civic Hall, Leeds on Wednesday, 27th March, 2019. Available online at: https://democracy.leeds.gov.uk/documents/g8292/Printed%20minutes%2027th-Mar-2019 %2013.00%20Council.pdf?T=1 (Accessed July 07, 2021).

Leeds City Council (2020). *Leeds Observatory – Population*. Available online at: https://observatory.leeds.gov.uk/population/ (accessed July 7, 2021).

Leeds City Council (2022). Local Plan Update. Local Plan Update. Available online at: https://www.leeds.gov.uk/lpu (accessed January 10, 2023).

Lockwood, M., Mitchell, C., and Hoggett, R. (2020). Incumbent lobbying as a barrier to forward-looking regulation: The case of demand-side response in the GB capacity market for electricity. *Ener. Policy* 140:111426. doi: 10.1016/j.enpol.2020.111426

Lowes, R., Rosenow, J., Qadrdan, M., and Wu, J. (2020a). Hot stuff: Research and policy principles for heat decarbonisation through smart electrification. *Ener. Res. Soc. Sci.* 70:101735. doi: 10.1016/j.erss.2020.101735

Lowes, R., Woodman, B., and Fitch-Roy, O. (2019). Policy change, power and the development of Great Britain's Renewable Heat Incentive. *Ener. Policy* 131, 410–421. doi: 10.1016/j.enpol.2019.04.041

Lowes, R., Woodman, B., and Speirs, J. (2020b). Heating in Great Britain: an incumbent discourse coalition resists an electrifying future. *Environ. Innov. Soc. Trans.* 37, 1–17. doi: 10.1016/j.eist.2020.07.007

Lund, H., Østergaard, P. A., Nielsen, T. B., Werner, S., Thorsen, J. E., Gudmundsson, O., et al. (2021). Perspectives on fourth and fifth generation district heating. *Energy* 227:120520. doi: 10.1016/j.energy.2021.120520

Markard, J., Raven, R., and Truffer, B. (2012). Sustainability transitions: An emerging field of research and its prospects. *Res. Policy* 41, 955–967. doi: 10.1016/j.respol.2012.02.013

Martiskainen, M., Schot, J., and Sovacool, B. K. (2021). User innovation, niche construction and regime destabilization in heat pump transitions. *Environ. Innov. Soc. Trans.* 39, 119–140. doi: 10.1016/j.eist.2021.03.001

MHCLG (2015). Statement made by Mr Eric Pickles, Secretary of State for Communities and Local Government. Available online at: https://questions-statements.parliament.uk/written-statements/detail/2015-03-25/HCWS488 (accessed November 28, 2021).

MHCLG (2018). Conservation of fuel and power: Approved Document L. 2013 edition with 2016 amendments. GOV.UK. Available online at: https://www.gov.uk/ government/publications/conservation-of-fuel-and-power-approved-document-l (accessed November 28, 2021).

MHCLG (2019). National Planning Policy Guidance: Climate change. GOV.UK. Available online at: https://www.gov.uk/guidance/climate-change (accessed November 28, 2021).

MHCLG (2020). Land value estimates for policy appraisal 2019. GOV.UK. Available online at: https://www.gov.uk/government/publications/land-value-estimates-for-policy-appraisal-2019 (accessed March 13, 2021).

MHCLG (2021a). National Planning Policy Framework. HM Government.

MHCLG (2021b). English Housing Survey, 2019 to 2020: Energy [WWW Document]. GOV.UK. Available online at: https://www.gov.uk/government/statistics/englishhousing-survey-2019-to-2020-energy (Accessed March 5, 2023).

National Grid (2018). *Future Energy Scenarios*. London: National Grid. Available online at: https://www.nationalgrideso.com/document/169921/download (Accessed March 07, 2020).

Nesta (2022). Could New Ways of Paying For Heat Help Reduce Household Carbon Emissions? London: National Endowment for Science, Technology and the Arts. Ofgem (2021a). Easy guide to shared ground loops for the Non-Domestic Renewable Heat Incentive (RHI), Non-domestic renewable heat incentive. Office for Gas and Electricity Markets.

Ofgem (2021b). NDRHI Closure. Ofgem. Available online at: https://www.ofgem. gov.uk/environmental-programmes/non-domestic-rhi/about-non-domestic-rhi/ ndrhi-closure (accessed March 26, 2021).

Ofgem (2021c). Non-Domestic Renewable Heat Incentive (RHI) - NDRHI Closure | Ofgem. Available online at: https://www.ofgem.gov.uk/environmental-andsocial-schemes/non-domestic-renewable-heat-incentive-rhi/ndrhi-closure (accessed February 15, 2024).

Ofgem (2022). Boiler Upgrade Scheme (BUS). Available online at: https://www. ofgem.gov.uk/environmental-and-social-schemes/boiler-upgrade-scheme-bus (accessed March 6, 2023).

ONS (2021). Estimates of the population for the UK, England and Wales, Scotland and Northern Ireland - Office for National Statistics. Available online at: https://www.ons.gov.uk/ peoplepopulationandcommunity/populationandmigration/populationestimates/ datasets/populationestimatesforukenglandandwalesscotlandandnorthernireland (accessed July 7, 2021).

Rankl, F. (2023). Overview of the planning system (England). London: House of Commons Library. Available online at: https://commonslibrary.parliament.uk/planning-in-england/ (accessed November 26, 2023)

Rankl, F. (2024). *Planning Appeals (England) (No. SN 6790), Research Briefing.* London: House of Commons Library. Available online at: https://researchbriefings. files.parliament.uk/documents/SN06790/SN06790.pdf (accessed April 11, 2024).

Raven, R., Schot, J., and Berkhout, F. (2012). Space and scale in socio-technical transitions. *Environ. Innov. Soc. Trans.* 4, 63–78. doi: 10.1016/j.eist.2012.08.001

Rees, S. J. (2019). Looking forward to UK low-carbon heating. *Build. Serv. Eng. Res. Technol.* 40, 665–668. doi: 10.1177/0143624419877750

Rip, A., and Kemp, R. (1998). "Technological change," in *Human choice and climate change: Vol. II, Resources and Technology*, eds. S. Rayner, E. L. Malone (Columbus, OH: Battelle Press).

Ritchie, H., Rosado, P., and Roser, M. (2023).  $CO_2$  and Greenhouse Gas Emissions. Our World in Data.

Roelich, K., Bale, C. S. E., Turner, B., and Neall, R. (2018). Institutional pathways to municipal energy companies in the UK: Realising co-benefits to mitigate climate change in cities. *J. Clean. Prod.* 182, 727–736. doi: 10.1016/j.jclepro.2018. 02.002

Rosenow, J., Lowes, R., Broad, O., Hawker, G., Wu, J., Qadrdan, M., et al. (2020). *The pathway to net zero heating in the UK.* UK Energy Research Centre Energy Data Centre (UKERC EDC).

Rosenow, J., and Thomas, S. (2020). UK heating plan still means 120 gas boilers installed for every low-carbon system. Energy Post. Available online at: https://energypost.eu/uk-heating-plan-still-means-120-gas-boilers-installed-for-every-low-carbon-system/ (accessed August 28, 2022).

Schwanen, T. (2015). The bumpy road toward low-energy urban mobility: case studies from two UK cities. *Sustainability* 7, 7086–7111. doi: 10.3390/su706 7086

Shabha, G., Barber, F., and Laycock, P. (2021). A qualitative assessment of the impact of smart homes and environmentally beneficial technologies on the UK 2050 net-zero carbon emission target. *Smart Sustain. Built Environ.* 12, 341–360. doi: 10.1108/SASBE-07-2021-0112

Simpson, K., Janda, K. B., and Owen, A. (2020). Preparing 'middle actors' to deliver zero-carbon building transitions. *Build. Cities* 1, 610–624. doi: 10.5334/bc.53

Slorach, P. C., and Stamford, L. (2021). Net zero in the heating sector: technological options and environmental sustainability from now to 2050. *Energy Conver. Manage.* 230:113838. doi: 10.1016/j.enconman.2021.113838

Sperling, K., Hvelplund, F., and Mathiesen, B. V. (2011). Centralisation and decentralisation in strategic municipal energy planning in Denmark. *Ener. Policy* 39, 1338–1351. doi: 10.1016/j.enpol.2010.12.006

Sullivan, R., Gouldson, A., and Webber, P. (2013). Funding low carbon cities: local perspectives on opportunities and risks. *Clim. Policy* 13, 514–529. doi: 10.1080/14693062.2012.745113

Swilling, M., Musango, J., and Wakeford, J. (2016). Developmental states and sustainability transitions: prospects of a just transition in South Africa. J. Environ. Policy Plann. 18, 650–672. doi: 10.1080/1523908X.2015.1107716

Taylor, P. G., Bolton, R., Stone, D., and Upham, P. (2013). Developing pathways for energy storage in the UK using a coevolutionary framework. *Energ. Policy* 63, 230–243. doi: 10.1016/j.enpol.2013.08.070

The Climate Change Act (2008). *The Climate Change Act 2008* (2050 *Target Amendment*) Order 2019, 2019), No. 1056. Available online at: https://www.legislation.gov.uk/uksi/2019/1056/contents/made (Accessed March 14, 2021).

Thellufsen, J. Z., and Lund, H. (2016). Roles of local and national energy systems in the integration of renewable energy. *Appl. Energy* 183, 419-429. doi: 10.1016/j.apenergy.2016.09.005

Tingey, M., and Webb, J. (2020). Governance institutions and prospects for local energy innovation: laggards and leaders among UK local authorities. *Energy Policy* 138:111211. doi: 10.1016/j.enpol.2019.111211

Tomaney, J., and Colomb, C. (2018). *Devolution and Planning, in: Planning Practice*. London: Routledge. doi: 10.4324/9781351203319-2

Torrens, J., Johnstone, P., and Schot, J. (2018). Unpacking the formation of favourable environments for urban experimentation: the case of the bristol energy scene. *Sustainability* 10:879. doi: 10.3390/su10030879

Truffer, B., Murphy, J. T., and Raven, R. (2015). The geography of sustainability transitions: Contours of an emerging theme. *Environ. Innov. Soc. Trans.* 17, 63–72. doi: 10.1016/j.eist.2015.07.004

Upton, W. (2019). What is the purpose of planning policy? Reflections on the revised national planning policy framework 2018. J. Environ. Law 31, 135–149. doi: 10.1093/jel/eq2005

Verbong, G., Christiaens, W., Raven, R., and Balkema, A. (2010). Strategic Niche Management in an unstable regime: Biomass gasification in India. *Environ. Sci. Policy* 13, 272–281. doi: 10.1016/j.envsci.2010. 01.004

Webb, J., Hawkey, D., and Tingey, M. (2016). Governing cities for sustainable energy: the UK case. *Cities, Cities, Ener. Clim. Chan. Mitigat.* 54, 28–35. doi: 10.1016/j.cities.2015.10.014

Webb, J., Tingey, M., and Hawkey, D. (2017). What We Know about Local Authority Engagement in UK Energy Systems: Ambitions, Activities, Business Structures and Ways Forward. London, UK: UK Energy Research Centre and Loughborough, Energy Technologies Institute.

Yin, R. K. (2018). Case Study Research and Applications: Design and Methods, Sixth. ed. Thousand Oaks, CA: SAGE Publications, Inc.